Reviewer #1:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: great job
Response:
Thank you for your comment. We have invited a professional native English speaker to help revise the language of the article and we believe we can meet your requirements.

Reviewer #2:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision
Specific Comments to Authors: Good work done by the authors by reporting an usual case of pheochromocytoma-induced atypical Takotsubo cardiomyopathy which can present with chest pain, the common presenting symptom for myocardial infection. I have marked few short coming in the manuscript which has been marked in the returned file. The grammar needs further polishing. The authors are requested to report on the reported incidence on pheochromocytoma-induced atypical Takotsubo cardiomyopathy. A small comment on how the patient was prepare for surgery. Also, what type of procedure was done(open versus laparoscopic vs robotic). what precaution and measures were taken operatively. comment on histopathological features of specimen. post-op medication. How frequently patient was followed. on each follow up what investigations were done.
Response:
Thank you for your comment.
1. We have invited a professional native English speaker to help revise the language of the article and we believe we can meet your requirements.
2. We have added to the epidemiological status of the disease, written in line 167-169 of the manuscript.
3. We have added to the patient's preparation for surgery and type of surgery, written in line 129-132 of the manuscript.
4. After the patient was admitted to the hospital, we took effective preventive measures and treated symptomatically to prevent and control the patient's recurrence of malignant hypertension, myocardial infarction and other adverse events, and performed surgery with the application of doxazosin, written in line 125-132 of the manuscript.
5. Regarding the histopathological characteristics of the specimens, we describe them in the Legend section of Figure 4, and the diagnosis was written in line 219-225 of the manuscript.
6. We added the postoperative medications, the patient was doing well postoperatively without any significant discomfort, and only lipid-lowering therapy was administered. Also, we followed her up after 1 year and the patient did not have similar symptoms again, and her ECG and echocardiogram were normal, but unfortunately, no imaging data were collected at the time of follow-up. These were written in line 134-139 of the manuscript.

Reviewer #3:
Scientific Quality: Grade D (Fair)
Language Quality: Grade B (Minor language polishing)
Conclusion: Major revision
Specific Comments to Authors: Dear authors: Atypical Takotsubo cardiomyopathy presenting as acute coronary
syndromes: a case report and literature review could be an interesting case to report, as you alluded in this article. The most parts of the manuscript have an acceptable spelling, grammar and syntax. However, before it becomes publishable, it still requires some improvement. Here are my comments:

1. The abstract must be included of background, case summary, and conclusion. You must arrange them by this maxim.
2. There are no keywords in your manuscript. For determining the correct and most appropriate keywords, you can use Medical Subject Headings (MeSH) or Google Keyword Planner.
3. You did not allude Core Tip in your manuscript.
4. The introduction is very short. The constructs and concepts in the introduction section are poor-organized and incomplete. Include more general and specific background in this section.
5. If there are any kinds of Laboratory examinations, you must mention them in table format with precise figures.
6. The case presentation must be included of Chief complaints, History of past illness, Physical examination, etc. You can download one of the accepted articles in the journal and see the format.
7. Most bibliographic citations which been used are more than 5 years old and obsolete. The authors must update and arrange the bibliography. Important note: the manuscript has not been written with journal's standards.

Response:
Thank you for your comment.

1. We have invited a professional native English speaker to help revise the language of the article and we believe we can meet your requirements.
2. We apologize for not adhering to the journal standards, we have reformatted the manuscript and updated the references, hopefully to your satisfaction.
3. We noticed that the introduction was too simple and made additions, hopefully to your satisfaction.

Thank you for your specific suggestions. In summary, we have made major revisions to the manuscript and have explained every point you raised, so please review it again.

Sincerely,

Reviewer #4:

Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision

Specific Comments to Authors: The authors describe Atypical Takotsubo cardiomyopathy presenting as acute coronary syndromes which is quite interesting. Overall, the article is well written. Some points need to be modified however: 1. Mention the doses of drugs 2. It doesn't provide sufficient content to make a literature review. However, case report is sufficient.

Response:
Thank you for your comment.

1. We have invited a professional native English speaker to help revise the language of the article and we believe we can meet your requirements.
2. As per your suggestion, we have reorganized the medication record and added to the discussion, hopefully to your satisfaction.